

## Author Response Letter to Reviewers' Comments

**Name of Journal:** *World Journal of Methodology*

**Manuscript NO:** 40802

**Manuscript Type:** Editorial

**Manuscript Title:** Precision medicine allergy immunoassay methods for assessing IgE sensitization to aeroallergen molecules

**Authors:** Florin-Dan Popescu and Mariana Vieru

**Correspondence to:** Florin-Dan Popescu, MD, PhD, Associate Professor, Department of Allergology, "Carol Davila" University of Medicine and Pharmacy, "Nicolae Malaxa" Clinical Hospital, Department of Allergology and Clinical Immunology, Bucharest 022441, Sos Vergului 12, Sector 2, Romania, European Union. florindanpopescu@ymail.com

**Reviewer Codes:** 02510721, 02520738, and 03723709

**Scientific Editor:** Li-Jun Cui

**Editorial Office Science Editor:** Ying Dou

Dear Editors,

We, the authors, appreciate your care and valuable suggestions for our manuscript mentioned above, which is an invited editorial.

I have attached the manuscript revised according to the reviewers and editor comments. The references were revised according to Science Editor comments.

Point-by-point responses to the reviewers' comments are listed down in this letter. I hope these changes meet your expectations.

Best regards,

Florin-Dan Popescu

Corresponding author. MD, PhD, Assoc. Professor, Department of Allergology, "Carol Davila" University of Medicine and Pharmacy, Bucharest, Romania.

## **Author's Response to the Reviewer's Comment**

**Reviewer's code:** 02510721

### **Reviewer's Comment:**

This editorial, on the complex problem of assessment of IgE sensitization by allergy immunoassay methods, is very complete and detailed. The references are updated. Overall this is an excellent work. I propose only one suggestion: for a better understanding of the matter can be useful review the manuscript completely, using shorter sentences. In fact many sentences are very long and difficult to follow and understand.

### **Authors' Response:**

Dear reviewer,

Thank you so much for your overall evaluation of our manuscript, and for valuable comments and suggestions. Please allow us to segment your comment into two points for the response.

### **Point-by-point responses:**

□ **Comment 1: The complex problem of assessment of IgE sensitization by allergy immunoassay methods, is very complete and detailed. The references are updated. Overall this is an excellent work.**

o **Response:** Thank you very much for assessing our editorial manuscript as an excellent work, and very good for its scientific quality. There is indeed a great need for very complete, integrated and detailed publications regarding all specific IgE immunoassays (singleplex, multiparameter or multiplex), with updated bibliographic references, because new generation methods are recently introduced in allergy research and clinical practice.

□ **Comment 2: I propose only one suggestion: for a better understanding of the matter can be useful review the manuscript completely, using shorter sentences. In fact many sentences are very long and difficult to follow and understand.**

o **Response:** You are right that, even if there are no language errors and only minor language polishing is needed, shorter sentences are better for easier reading and understanding. I completely revised the manuscript, as you suggested, making shorter sentences in all paragraphs, as possible, without other changes in information, in order not to modify the meaning of the phrases of such a very complex technical text. Moreover, comprehensive editing processes were applied on our manuscript in order to achieve Grade A: priority publishing, with no language polishing required after editing.

**Reviewer's code:** 02520738

**Reviewer's Comment:** To: Editorial Board World Journal of Methodology  
Title: "Precision medicine allergy immunoassay methods for assessing IgE sensitization to aeroallergen molecules" Dear Editor, I read this manuscript and I think that the paper is good and well written.

**Authors' Response:**

Dear reviewer,

Thank you for your valuable time spent on our manuscript and for considering it well written, giving a Grade A: Priority publishing. In addition, thank you for assessing our work as Grade A: Excellent for its scientific quality.

**Reviewer's code:** 03723709

**Reviewer's Comment:**

The author presented a comprehensive interesting editorial about "Precision medicine allergy immunoassay methods for assessing IgE sensitization to aeroallergen molecules". It is a very interesting editorial indeed, that I enjoyed reading it. Good work.

**Authors' Response:**

Dear reviewer,

Thank you for your kind overall evaluation and your comments regarding the recent specialists interest on the molecular-based allergy diagnosis for the *in vitro* evaluation of patients IgE sensitization profiles at the molecular level. Thank you for assessing our manuscript with Grade A: Priority publishing, and for your appreciation that you enjoyed reading it.